



# **Geophysical Research Letters**

# **RESEARCH LETTER**

10.1029/2018GL079593

#### **Kev Points:**

- Hadley cell expansion trends in early-generation reanalyses are larger than those in modern reanalyses
- · The majority of these larger trends are impacted by mass nonconservation in the mean meridional circulation
- When this is corrected, Hadley cell expansion trends in early-generation reanalyses converge toward those in modern reanalyses

#### **Supporting Information:**

• Supporting Information S1

#### **Correspondence to:**

N A Davis nicholas.davis@noaa.gov

#### Citation:

Davis, N. A., & Davis, S. M. (2018). Reconciling Hadley cell expansion trend estimates in reanalyses. Geophysical Research Letters, 45, 11,439-11,446. https://doi.org/10.1029/2018GL079593

Received 11 JUL 2018 Accepted 3 OCT 2018 Accepted article online 8 OCT 2018 Published online 23 OCT 2018

# **Reconciling Hadley Cell Expansion Trend Estimates in Reanalyses**

# N. A. Davis<sup>1</sup> 🕞 and S. M. Davis<sup>1</sup>

<sup>1</sup>University of Colorado Cooperative Institute for Research in Environmental Sciences (CIRES) at the NOAA Earth System Research Laboratory (ESRL) Chemical Sciences Division, Boulder, CO, USA

Abstract Numerous studies have concluded that historical Hadley cell expansion simulated in reanalyses is much larger than the future expansion predicted by climate model simulations. Is Hadley cell expansion too weak in climate models, or are the trends in reanalyses spuriously large? This study shows that the mean meridional circulation in reanalyses generally does not conserve mass. The mass imbalance projects onto trends in the Hadley cell edge latitudes by modifying both the mean and anomalous circulation. In correcting for the imbalance, the majority of Hadley cell expansion trends in early-generation reanalyses in both hemispheres are revised to be smaller in magnitude, bringing them into closer agreement with the trends in modern reanalyses and climate models. While the methodology presented here is statistical in nature, it produces quantitatively similar results to a more sophisticated mass budget correction method.

Plain Language Summary Reanalyses are weather forecast models that use observations to constrain hindcasts of the past evolution of Earth's atmosphere. They are one tool used to study the Hadley cells, two large circulation cells in the tropics with far-reaching impacts on Earth's climate. The amount the Hadley cells expanded in older reanalysis products was much larger than what was simulated in climate models. However, some of this expansion is due to the violation of a basic physical law. When this is corrected, the expansion in older reanalyses is closer to the expansion in more modern reanalyses. It is possible that some of the supposed discrepancy between models and reanalyses was due to these sorts of problems in the older reanalyses.

# 1. Introduction

While differential solar heating of Earth's surface leads to the meridional variation of temperature, the Hadley circulation shapes the climate of Earth's tropical belt by redistributing heat and water vapor. Deserts and stratocumulus decks form at the belt's northern and southern edges on the eastern side of ocean basins, where subsidence dries and stabilizes the atmosphere. Water vapor in these subtropical dry zones is converged onto the equator by the trade winds, where it heats the air through condensation and drives the upward motion in the Hadley cells. Divergent flow aloft carries this warm air poleward, where it can be tapped by eddies and transported into the midlatitudes (Trenberth & Stepaniak, 2003).

Models predict that the Hadley cell edges will expand poleward in response to radiative forcings (Davis et al., 2016; Grise & Polvani, 2016; Hu et al., 2013; Lu et al., 2007), and there is emerging evidence that such changes may have already occurred (Choi et al., 2014; Davis & Rosenlof, 2012; Lucas et al., 2014; Seidel et al., 2008; Tselioudis et al., 2016). With strong gradients in temperature, humidity, precipitation, and wind at the belt's edges, any poleward shift in the circulation could project strongly onto the surface climate of Earth's most populous latitudes (Birner et al., 2014). As there are insufficient observations to study the mean meridional circulation directly (Waliser et al., 1999), reanalyses have been used to estimate past trends in the Hadley cell edge latitudes (Allen et al., 2012; Davis & Birner, 2013; Hu & Fu, 2007; Johanson & Fu, 2009; Nguyen et al., 2013; Stachnik & Schumacher, 2011). These studies have shown a statistically significant poleward movement of the Hadley cell edge latitudes in both hemispheres over the last ~30 years, and a number of studies have concluded that climate model trends are much weaker than those in reanalyses (Johanson & Fu, 2009; Quan et al., 2014). However, a recent study using modern reanalyses suggests that the disagreements, if they exist, are potentially small (Davis & Birner, 2017), especially in light of the magnitude of internal variability (Garfinkel et al., 2015).

©2018. American Geophysical Union. All Rights Reserved.

This paper addresses the extent to which these seemingly conflicting conclusions are driven by the use of early-generation versus modern reanalyses (defined in the next section), and why the results might be

reanalysis dependent. It is well known that reanalyses may have spurious trends, as the combination of model biases and inhomogeneities in the observing/assimilation system may lead to unphysical jumps and violations of mass/energy conservation (Bengtsson et al., 2004; Berrisford et al., 2011; Long et al., 2017; Quan et al., 2014). Here we explore the possibility that Hadley cell expansion trends in early-generation reanalyses are spuriously large and that the modern reanalyses more accurately reflect past changes in Hadley cell width. To the extent possible we want to know whether inhomogeneities in reanalysis mass conservation relate to spurious changes in the mean meridional circulation, and what effect this might have on Hadley cell edge latitude time series and their derived trends. We will also examine possible corrections for mass non-conservation and how these corrections relate to the previously reported discrepancies among reanalysis trend estimates of Hadley cell expansion.

# 2. Data

We use vertical and meridional winds from nine reanalyses: the National Centers for Environmental Prediction (NCEP)/National Center for Atmospheric Research (NCAR) reanalysis (Kalnay et al., 1996), the NCEP/Department of Energy (DOE) reanalysis (Kanamitsu et al., 2002), and the Climate Forecast System Reanalysis Version 2, hereafter *CFSR* (Saha et al., 2010); the European Centre for Medium-Range Weather Forecasting (ECMWF) 40-year Reanalysis, hereafter *ERA-40* (Uppala et al., 2006), and the ECMWF Interim Reanalysis, hereafter *ERA-Interim* (Dee et al., 2011); the Japanese Meteorological Agency 25-year Reanalysis, hereafter *JRA55* (Kobayashi et al., 2007), and the Japanese Meteorological Agency 55-year Reanalysis, hereafter *JRA55* (Kobayashi et al., 2015); and the National Aeronautics and Space Administration (NASA) Modern Era Retrospective Analysis for Research and Applications (Rienecker et al., 2011), hereafter *MERRA*, and the Modern Era Retrospective Analysis for Research and Applications 2 (Gelaro et al., 2017), hereafter *MERRA*2. We use the assimilation, or *asm* product, rather than the analysis, or *ana* product, from MERRA and MERRA2 (the resulting trends are sensitive to the output product [Garfinkel et al., 2015]). For further information on these reanalyses, see Table 1 of Fujiwara et al. (2017).

JRA55, MERRA2, CFSR, and ERA-Interim are designated here as modern reanalyses, as they represent each agency's most recent reanalysis product. NCEP/NCAR, NCEP/DOE, ERA-40, JRA25, and MERRA are referred to here as early-generation reanalyses. We recognize this distinction is somewhat arbitrary for MERRA, as MERRA is just as modern as ERA-Interim. Nevertheless, MERRA is NASA's first-generation reanalysis product and has since been supplanted by MERRA2, so it is no longer NASA's state-of-the-art product.

We also use 36 Coupled Model Intercomparison Project—Phase 5 (CMIP5; Taylor et al., 2012) Historical and 23 CMIP5 RCP8.5 simulations, with the models listed in the supporting information.

Our period of analysis is 1980 through 2009, though ERA-40 ends in 2001 and JRA25 ends in 2004.

## 3. Methods

The mean meridional stream function (MMS) is the standard variable for examining the Hadley cells. It is defined as the meridional mass flux between a given level and the top of the atmosphere, that is,

$$\Psi(\boldsymbol{p},\boldsymbol{\phi}) = \frac{2\pi a \cos \boldsymbol{\phi}}{g} \int_{\boldsymbol{p}}^{0} [\boldsymbol{v}] \mathrm{d}\boldsymbol{p}, \tag{1}$$

where  $\Psi(p, \phi)$  is the MMS at pressure *p* and latitude  $\phi$ , *a* is the radius of the Earth, *g* is the gravitational acceleration, and [*v*] is the zonal mean meridional wind where brackets indicate the zonal mean. We define the Hadley cell edge latitudes as the latitudes of the zero crossing of the MMS at 500 hPa in each hemisphere poleward of each hemisphere's maximum absolute MMS.

For a meridional circulation that conserves mass, the vertical velocity is a function of the meridional gradient of the MMS and is given by

$$[\omega](\mathbf{p},\phi) = -\frac{g}{2\pi a^2 \cos\phi} \frac{\partial \Psi(\mathbf{p},\phi)}{\partial \phi}$$
(2)

where  $[\omega]$  is the zonal mean vertical pressure velocity. For the purposes of this study, it is worth emphasizing that by construction the stream function is able to simultaneously describe the vertical and meridional flow





**Figure 1.** Twenty-four-month running mean residual mass flux at 500 hPa from the nine reanalyses. Dots indicate data stream transitions; vertical bars indicate the assimilation of new satellite observations, with vertical gray bars indicating the particular satellite mission. Dates of both data stream and satellite transitions and color scheme from Fujiwara et al. (2017). NCEP/NCAR, NCEP/DOE, and CFSR scaled by factors of 1/20, 1/2, and 1/10, respectively. For the AMSU A transition, MERRA and MERRA2 begin to assimilate both AMSU A and B. For the COSMIC and AIRS transition, CFSR and ERA-Interim begin to assimilate AIRS, while MERRA2 begins to assimilate COSMIC.

only if mass is conserved. Without this constraint, two-dimensional flow cannot be described by a single variable.

As isobars in a hydrostatic fluid are surfaces of constant vertically integrated mass, the globally integrated vertical mass flux on an isobar must be 0 (the mass flux is equivalent to the vertical pressure velocity divided by *g*). Otherwise, mass conservation is violated. One might be tempted to use the vertical velocity derived from the MMS to test whether the MMS conserves mass. But by construction, the vertical pressure velocity derived from the MMS will be exactly what is required to conserve mass given any meridional wind field.

This presents a conundrum, as we cannot use the MMS itself to test whether the MMS conserves mass. Instead, we examine the global mean of the vertical mass flux at 500 hPa in the reanalyses. As it should be 0 if the circulation conserves mass globally, we refer to this quantity as the global mean *mass flux residual*. Five hundred hectopascals is chosen as it is most relevant for the Hadley cell edge latitude metric. In models the vertical pressure velocity is generally diagnosed from the surface pressure tendency and the divergence of the horizontal winds (e.g., it is the vertical mass flux necessary to close the column mass budget). It is therefore at minimum as well constrained as the zonal mean meridional wind.

It may be the case that a reanalysis conserves mass, at least as measured by the global mean surface pressure (e.g., Gelaro et al., 2017) but does not conserve mass locally as measured by the global mean mass flux residual at 500 hPa. Local inconsistencies between the vertical and horizontal circulation may arise because of the analysis tendencies imposed on the horizontal winds and the geopotential. Regardless of whether the reanalysis conserves global mean mass or not, any local mass nonconservation in the circulation could project onto variability and trends in the Hadley cell edge latitudes.

# 4. Results

We first examine the 2-year moving average monthly mass flux residual at 500 hPa (Figure 1; the raw time series are shown in Figure S1 in the supporting information). Here positive values indicate a net downward mass flux—the same sign as the pressure velocity. For reference, the values shown here range from 0.5% to 3.0% of the root-mean-square 500 hPa mass flux in the reanalyses (or approximately on the order of parts per hundred of the atmospheric mass on daily timescales). In many cases, the variability in mass balance across reanalyses is uncorrelated, which could be taken as characteristic of a nonphysical process.

In addition to the mass flux residual, Figure 1 notes data stream transitions and satellite transitions to aid in the interpretation of variability in the mass flux residual. During a satellite transition, a reanalysis begins to assimilate a new set of satellite observations with new biases and coverage statistics. These changes to the observing system can introduce artificial changes into the simulated climate. During a data stream transition, two different reanalysis simulations are stitched together to a form a complete record of the climate system, which can create similar discontinuities. The satellite transitions indicated include the Special Sensor Microwave/Imager (SSM/I), Advanced Microwave Sounding Unit (AMSU) A and B, the Atmospheric Infrared Sounder (AIRS), and the Challenging Minisatellite Payload (CHAMP) and Constellation Observing System for Meteorology Ionosphere and Climate (COSMIC) Global Positioning System-Radio Occultation sytems. Further information, including the dates of most of these transitions, can be found in Fujiwara et al. (2017). We will comment on the coincidence of these transitions and peculiar behavior in the mass flux residual, but we do not assess the plausibility of these transitions in causing such behavior.

ERA-Interim and ERA40 are nearly indistinguishable, in part because their mass flux residual is essentially 0 throughout the data record. There does appear to be a minor trend in the mass flux residual in ERA-Interim.

MERRA transitions from a positive to near-zero mass flux residual after the SSM/I transition and then to a negative mass flux residual during a data stream transition and during the period when AMSU A, AMSU B,

Atmospheric Infrared Sounder, and COSMIC began to be assimilated. MERRA2 is clearly an improvement over MERRA, though it exhibits a long-term drift over the record, in part due to a reduction in the mass flux residual after the SSM/I transition.

Improvement is also evident between JRA25 and JRA55. JRA25 has a positive mass flux residual in the late 1980s, before a data stream transition and when SSM/I began to be assimilated. Later, it transitions from near zero to negative after it began to assimilate AMSU data. In contrast, JRA55 maintains relative mass balance throughout the modern era, with minor variability around the SSM/I and COSMIC transitions. Even in the early 2000s, when there were substantial changes to the observing system, JRA55 exhibits little variability.

NCEP/NCAR, which must be scaled by a factor of 1/20 to fit within the plot axes, exhibits several abrupt shifts in mass flux residual after the year 2006 (see also Figure S1). Perhaps because of the improvements in the representation of physical processes between NCEP/NCAR and NCEP/DOE, the mass flux residual in NCEP/DOE is smaller over most of the observational period. In the 2010s the mass flux begins to rapidly diverge toward larger values. However, the connection between satellite transitions and unphysical behavior in NCEP/NCAR and NCEP/DOE is not as straightforward to assess as it is in other reanalyses. NCEP/NCAR and NCEP/DOE assimilate temperature retrievals, unlike the other reanalyses that assimilate radiance measurements. While CFSR does not have such abrupt changes in its mass flux residual, it does have the largest root-mean-square mass flux residual with a substantial long-term trend. Like MERRA, it displays a rapid shift from positive to negative values in the late 1990s and early 2000s, coincident with the AMSU transitions. However, it appears that CHAMP may reanchor CFSR to a lower mass flux residual shortly thereafter.

We have examined the impact of horizontal resolution on the mass flux residual to test whether the mass flux residual is a numerical artifact (Figure S2). Here we examine the root-mean-square mass flux residual, rather than the mean or standard deviation, as we are interested in quantifying representative values of the mass flux residual regardless of sign. In subsampling the reanalyses to coarser resolutions, we find that the root-mean-square mass flux residual does not increase. This suggests that the mass flux residuals are not artifacts of insufficient horizontal resolution but are instead large-scale features of the modeled climate in reanalyses. As a measure of calculation uncertainty, we also calculated the root-mean-square mass flux residual in the JRA55-AMIP simulation and in the CMIP5 Historical simulations. In these simulations, the underlying model is not coupled to an assimilation system and is expected to better conserve mass. The CFSR, NCEP/NCAR, NCEP/DOE, and MERRA reanalyses have mass flux residuals greater than the estimates from the CMIP5 and JRA55-AMIP simulations, suggesting that their mass flux residuals are likely to be manifestations of mass non-conservation. On the other hand, the mass flux residuals in the remaining reanalyses are indistinguishable from our estimate of calculation error.

To gain some insight into the impact of this mass flux residual on the circulation, we regress the MMS onto the monthly standardized, detrended mass flux residual (i.e., with zero mean and standard deviation of 1) in each reanalysis using a standard linear least squares regression (Figure 2). We do not assume that the climatological circulation in the reanalyses conserves mass, so we do not deseasonalize the index. For both the NCEP/NCAR and NCEP/DOE regressions, we exclude all values of the mass flux residual past 2006 when the abrupt changes occur. The regressions are marginally different if the regression is performed through 2009 (see Figure S3). For NCEP/NCAR, the regression pattern appears more similar to the others if one scales the regression by more than one standard deviation, for example, representative of the abrupt shifts (also see Figure S3).

In general, the mass flux projection is dominated by a single negative cross-equatorial cell in the tropics of the same sign as the Southern Hemisphere winter cell. In most reanalyses the projections also have two positive-signed cells along the zero contour of the MMS of the same sign as the Northern Hemisphere Hadley cell (Figure 2). These anomalies project oppositely onto the Hadley cell edge latitudes in each hemisphere. For example, given a positive mass flux residual the regressions indicate that the residual is associated with an anomalously poleward Northern Hemisphere Hadley cell edge and an anomalously equatorward Southern Hemisphere Hadley cell edge. CFSR is an outlier as it indicates opposite-signed impacts to the two hemispheres. ERA-Interim is also an outlier with no projection onto the Southern Hemisphere subtropical circulation. While the mass flux residual is small in ERA-Interim, its projection is concentrated at the Northern Hemisphere Hadley cell edge. Crucially, the projection in ERA-Interim for a given standard deviation change in the mass flux residual is quantitatively similar to the other reanalyses. In every reanalysis except



### Projection of mass flux residual

**Figure 2.** The projection of the mass flux residual onto the mean meridional stream function from the nine reanalyses (shading). Also shown is the annual mean meridional stream function (black contours, positive values in solid, negative values in dashed, zero line in dotted, every  $30 \times 10^9$  kg/s) from each reanalysis. Gray shading indicates coefficients not statistically significant at the 95% confidence level. Projections (and mass flux residuals in Figure 1) for the MERRA2, ERA-Interim, and NCEP/DOE reanalyses multiplied by -1 for ease of comparison.

NCEP/NCAR and ERA-Interim, the projections are mostly statistically significant wherever the values are contoured.

As the mass flux residual projects onto the MMS, it is possible to correct for the impact of mass nonconservation on the MMS by removing the projection of the mass flux residual at each time step. We derive the projection of the monthly mean mass flux residual on the MMS for each month, remove this from the monthly mean MMS to construct the *corrected* monthly mean MMS, and average the corrected monthly mean MMS into the corrected annual mean MMS. While this correction is entirely statistical in nature, we will corroborate some of our results using the more rigorous mass correction methodology of Trenberth (1991).

As noted before, we do not deseasonalize the mass flux residual or MMS, and for the correction we also do not detrend, either, though the regression coefficients used to correct the MMS are based on the detrended regression. Trends in the Hadley cell edge latitude are not a function of MMS anomalies alone but instead the superposition of MMS anomalies and the climatological MMS. For different meridional gradients in the climatological MMS, a given MMS anomaly will produce different changes in the Hadley cell edge latitude. It is therefore important to consider the impact of the mass flux residual on both the MMS anomalies and climatology.

Our focus is the impact of this correction on trends in the Hadley cell edge latitudes calculated from the annual mean MMS. Past studies have generally focused on these annual mean trends as the edge latitudes are often poorly defined in the summer months on monthly timescales. For NCEP/NCAR and NCEP/DOE, we use the regression pattern from Figure 2, which was only calculated based on data through 2006, on the full time series through 2009.

Correcting for the mass flux residuals generally decreases Northern Hemisphere and Southern Hemisphere expansion trends, although the impact differs by reanalysis product and is less robust in the Southern Hemisphere (Figure 3). In the Northern Hemisphere, the substantial and statistically significant poleward



**Figure 3.** Raw and corrected poleward trends in the Hadley cell edge latitudes in the Northern (left) and Southern (right) Hemispheres in the annual mean. Whiskers indicate 95% confidence intervals using the effective degrees of freedom, or, for CMIP5, the range of model trends. Stars indicate that the raw and corrected trends in the given reanalysis are statistically significantly different at the 95% confidence level. Magenta circles indicate the poleward trends in the Hadley cell edges using the Trenberth (1991) mass correction method—see text. Note that ERA-40 trends are through 2001, JRA25 trends are through 2004, and all others are through 2009.

trends in NCEP/NCAR and NCEP/DOE are reduced by over 80% using the corrected data. In JRA55 and ERA-Interim, the already-low trends are halved, and in MERRA the slightly negative trend is corrected to be more positive. For all of these reanalyses, the raw and corrected trends are statistically significantly different at the 95% confidence level (assessed as in Lanzante, 2008). The trends are not significantly different in MERRA2 or CFSR. Somewhat surprisingly, there are no statistically significant trend revisions for either ERA-40 or JRA25 in either hemisphere. Both reanalyses have spurious trends and biases in their precipitation and temperature fields that may contribute to their expansion trends, regardless of whether the circulation is corrected to conserve mass (Kobayashi et al., 2015). Despite this, the average expansion trend in the Northern Hemisphere decreases from  $0.34 \pm 0.08^{\circ}$  to  $0.17 \pm 0.07^{\circ}$  per decade. Ignoring JRA25, the trend decreases from  $0.29 \pm 0.08^{\circ}$  to  $0.13 \pm 0.07^{\circ}$  per decade.

In the Southern Hemisphere, the poleward trends in NCEP/DOE and CFSR decrease, and the poleward trend in MERRA decreases so as to become equatorward. There are no other statistically significant changes. The average expansion trend decreases from  $0.32 \pm 0.08$  to  $0.19 \pm 0.07^{\circ}$  per decade, meaning that the average expansion trends are nearly identical in both hemispheres whether one examines the raw or corrected data. It is worth noting that in NCEP/DOE and ERA-40, the raw trend was statistically significant but the corrected trend is not. In the Northern Hemisphere, this is also true for ERA-Interim and NCEP/DOE. On the other hand, in JRA55, CFSR, and MERRA in the Southern Hemisphere the trends are revised to be significant. Of note is that the corrections to MERRA, MERRA2, JRA25, JRA55, and ERA-Interim are of opposite sign in the two hemispheres. In MERRA2 and JRA-55, the corrections are of nearly the same magnitude such that the correction in the total (north plus south) expansion trend is small.

Also displayed in Figure 3 are the Hadley cell expansion trends assessed from the MMS corrected using the methodology of Trenberth (1991). Trenberth (1991) describes a barotropic correction to the horizontal winds, which corrects for mass nonconservation in global analyses by taking into account the column-integrated mass budget residual. This scheme must be implemented on the highest time resolution reanalysis output available and so is costlier to perform than the statistical technique used here. We have therefore performed this correction using budget data readily available for three of the reanalyses from the National Center for Atmospheric Research and described in Trenberth and Fasullo (2018; see Acknowledgements). The Trenberth (1991) correction gives quantitatively similar results to our statistical correction method in the ERA-Interim, NCEP/NCAR, and CFSR reanalyses. For example, in both hemispheres the Trenberth (1991) method produces similarly drastic reductions in Hadley cell expansion trends in NCEP/NCAR, from significantly poleward to indistinguishable from 0. For ERA-Interim, the Trenberth (1991) correction is small but so is our statistical correction. The minor differences in the corrected trends may be the result of the different vertical structure of the two methods. The Trenberth (1991) method is barotropic and partitions the mass flux necessary to close the mass budget equally at all levels; the projection patterns we find tend to be

concentrated in the midtroposphere, where the Hadley cell edge latitude is calculated. In spite of this, the time-mean Trenberth (1991) correction to the MMS (Figure S4) is similar in structure to our correction for the three reanalyses (Figure 2; the projection of the mass flux residual is the correction).

Excepting ERA-Interim in the Southern Hemisphere, the modern reanalysis trends fall within the range of CMIP5 model trends. Here the CMIP5 model trends are taken from 1980 through 2009 by combining the Historical and RCP8.5 simulations for 23 models. Regarding the early-generation reanalyses, in the Southern Hemisphere the uncorrected trends in JRA25, ERA-40, and NCEP/DOE fall outside of the range of model trends, while in the Northern Hemisphere the uncorrected trends. Once corrected, the trends in the Southern Hemisphere in ERA-40 and NCEP/DOE and the trends in the Northern Hemisphere in NCEP/NCAR and NCEP/DOE fall within the range of model trends. Even after being corrected, the trends in JRA-25 and the Northern Hemisphere trend in ERA-40 remain outside of the range of model trends.

# 5. Conclusions

Mass is not conserved by the meridional circulation in reanalyses. Residual vertical mass fluxes project onto Hadley cell expansion trends, artificially inflating trends in both hemispheres. The characteristics of the mass flux residual suggest it is not simply a function of horizontal resolution, but instead a result of errors introduced by the assimilation system and/or the representation of physical processes within the reanalysis models. Detection of these errors and their correction was accomplished without the use of any observational constraints and with only monthly mean reanalysis output. As further evidence of the reliability of this method, the Trenberth (1991) correction technique provides similar results. However, the mechanisms that may drive the imbalance and its projection onto the circulation remain uncertain.

While the corrected expansion trends in early-generation reanalyses are generally closer to those in modern reanalyses and CMIP5 model simulations, some persistent differences remain. These remaining differences could be associated with other errors and biases that are beyond the scope of this analysis. Though modern reanalyses still have difficulty conserving mass, these errors tend to not project as strongly onto trends in the Hadley cell edge latitudes as they do in early-generation reanalyses. For these reasons it would seem prudent to exclude early-generation reanalyses from future analyses of derived circulation quantities such as the MMS, except as historical reference points to past literature.

Johanson and Fu (2009) found that Hadley cell expansion in reanalyses was larger than in the worst-case scenario climate model experiments. However, they used the NCEP/NCAR, NCEP/DOE, and ERA-40 reanalyses. Grise et al. (2018) have noted that the NCEP/NCAR reanalysis is an outlier in regards to its stream functionbased tropical expansion trends. Here we have additionally shown that in general a substantial fraction of the expansion trends in these particular reanalyses are associated with violations of mass conservation. Our correction technique revises these trends to be smaller and in closer agreement with model trends. Considering the substantial impact of natural variability on the observed rates of expansion (Allen & Kovilakam, 2017; Amaya et al., 2017), more work is warranted to determine if there truly are differences between the rates of expansion simulated in models and reanalyses.

## References

Allen, R. J., & Kovilakam, M. (2017). The role of natural climate variability in recent tropical expansion. Journal of Climate, 30(16), 6329–6350. https://doi.org/10.1175/JCLI-D-16-0735.1

Allen, R. J., Sherwood, S. C., Norris, J. R., & Zender, C. S. (2012). Recent Northern Hemisphere tropical expansion primarily driven by black carbon and tropospheric ozone. *Nature*, 485(7398), 350–354. https://doi.org/10.1038/nature11097

Amaya, D. J., Siler, N., Xie, S.-P., & Miller, A. J. (2017). The interplay of internal and forced modes of Hadley cell expansion: Lessons from the global warming hiatus. *Climate Dynamics*, 51(1-2), 305–319. https://doi.org/10.1007/s00382-017-3921-5

Bengtsson, L., Hagemann, S., & Hodges, K. I. (2004). Can climate trends be calculated from reanalysis data? *Journal of Geophysical Research*, *109*, D11111. https://doi.org/10.1029/2004JD004536

Berrisford, P., Kållberg, P., Kobayashi, S., Dee, D., Uppala, S., Simmons, A. J., et al. (2011). Atmospheric conservation properties in ERA-Interim. *Quarterly Journal of the Royal Meteorological Society*, 137(659), 1381–1399. https://doi.org/10.1002/qj.864

Birner, T., Davis, S. M., & Seidel, D. J. (2014). The changing width of Earth's tropical belt. *Physics Today*, 67(12), 38–44. https://doi.org/10.1063/ PT.3.2620

Choi, J., S-W Son, J. L., & Min, S.-K. (2014). Further observational evidence of Hadley cell widening in the Southern Hemisphere. *Geophysical Research Letters*, 41, 2590–2597. https://doi.org/10.1002/2014GL059426

Davis, N. A., & Birner, T. (2013). Seasonal to multi-decadal variability of the width of the tropical belt. Journal of Geophysical Research: Atmospheres, 118, 7773–7787. https://doi.org/10.1002/jgrd.50610

#### Acknowledgments

The authors thank Krzysztof Wargan and Ori Adam for their constructive comments, and we thank Kevin Grise for the CMIP5 Hadley cell expansion trends. Budget data for the Trenberth (1991) correction were obtained from NCAR's Climate Analysis Section at http://www. cgd.ucar.edu/cas/catalog/, with calculation details described in Trenberth and Fasullo (2018). MERRA and MERRA2 output is provided by NASA's Global Modeling and Assimilation Office at https://gmao.gsfc. nasa.gov/reanalysis/. NCEP hosts NCEP/NCAR at https://www.esrl.noaa. gov/psd/data/reanalysis/reanalysis. shtml and NCEP/DOE at https://www. esrl.noaa.gov/psd/data/gridded/data. ncep.reanalysis2.html. ERA-Interim and ERA-40 are available through the ECMWF at https://www.ecmwf.int/en/ forecasts/datasets/archive-datasets. NCAR's Research Data Archive hosts the output of the JRA25, JRA55, and CFSR reanalyses at https://rda.ucar.edu/. We acknowledge the World Climate Research Programme's Working Group on Coupled Modeling, which is responsible for CMIP, and we thank the climate modeling groups (listed in Table S1 in the supporting information) for producing and making available their model output. For CMIP the U.S. Department of Energy's Program for Climate Model Diagnosis and Intercomparison provides coordinating support and led development of software infrastructure in partnership with the Global Organization for Earth System Science Portals, CMIP5 data can be accessed at https://cmip.llnl.gov/ cmip5/data\_portal.html.

- Davis, N. A., & Birner, T. (2017). On the discrepancies in tropical belt expansion between reanalyses and climate models and among tropical belt width metrics. *Journal of Climate*, 30(4), 1211–1231. https://doi.org/10.1175/JCLI-D-16-0371.1
  - Davis, N. A., Seidel, D. J., Birner, T., Davis, S. M., & Tilmes, S. (2016). Changes in the width of the tropical belt due to simple radiative forcing changes in the GeoMIP simulations. Atmospheric Chemistry and Physics, 16(15), 10,083–10,095. https://doi.org/10.5194/acp-16-10083-2016
- Davis, S. M., & Rosenlof, K. H. (2012). A multidiagnostic intercomparison of tropical-width time series using reanalyses and satellite observations. Journal of Climate, 25, 1061–1078. https://doi.org/10.1175/JCLI-D-11-00127.1
- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137(656), 553–597. https://doi.org/ 10.1002/qj.828
- Fujiwara, M., Wright, J. S., Manney, G. L., Gray, L. J., Anstey, J., Birner, T., et al. (2017). Introduction to the SPARC Reanalysis Intercomparison Project (S-RIP) and overview of reanalysis systems. Atmospheric Chemistry and Physics, 17(2), 1417–1452. https://doi.org/10.5194/acp-17-1417-2017
- Garfinkel, C. I., Waugh, D. W., & Polvani, L. M. (2015). Recent Hadley cell expansion: The role of internal atmospheric variability in reconciling modeled and observed trends. *Geophysical Research Letters*, 42, 10,824–10,831. https://doi.org/10.1002/2015GL066942
- Gelaro, R., McCarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., et al. (2017). The Modern-Era Retrospective Analysis for Research and Applications, Version 2 (MERRA-2). Journal of Climate, 30(14), 5419–5454. https://doi.org/10.1175/JCLI-D-16-0758.1

Grise, K. M., Davis, S. M., Staten, P. W., & Adam, O. (2014). Regional and seasonal characteristics of the recent expansion of the tropics. Journal of Climate, 31, 6839–6856. https://doi.org/10.1175/JCLI-D-18-0060.1

Grise, K. M., & Polvani, L. M. (2016). Is climate sensitivity related to dynamical sensitivity? Journal of Geophysical Research: Atmospheres, 121, 5159–5176. https://doi.org/10.1002/2015JD024687

Hu, Y., & Fu, Q. (2007). Observed poleward expansion of the Hadley circulation since 1979. Atmospheric Chemistry and Physics, 7, 5229–5236. https://doi.org/10.5194/acp-7-5229-2007

- Hu, Y., Tao, L., & Liu, J. (2013). Poleward expansion of the Hadley circulation in CMIP5 simulations. Advances in Atmospheric Sciences, 30(3), 790–795. https://doi.org/10.1007/s00376-012-2187-4
- Johanson, C. M., & Fu, Q. (2009). Hadley cell widening: Model simulations versus observations. Journal of Climate, 22(10), 2713–2725. https://doi.org/10.1175/2008JCLI2620.1

Kalnay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L., et al. (1996). The NCEP/NCAR 40-year reanalysis project. Bulletin of the American Meteorological Society, 77(3), 437–471. https://doi.org/10.1175/1520-0477(1996)077<0437:TNYRP>2.0.CO;2

Kanamitsu, M., Ebisuzaki, W., Woollen, J., Yang, S.-K., Hnilo, J. J., Fiorino, M., & Potter, G. L. (2002). NCEP-DOE AMIP-II Reanalysis (R-2). Bulletin of the American Meteorological Society, 83(11), 1631–1644. https://doi.org/10.1175/BAMS-83-11-1631

Kobayashi, S., Ota, Y., Harada, Y., Ebita, A., Moriya, M., Onoda, H., et al. (2015). The JRA55 reanalysis: General specifications and basic characteristics. Journal of the Meteorological Society of Japan, 93(1), 5–48. https://doi.org/10.2151/jmsj.2015-001

Lanzante, J. R. (2005). A cautionary note on the use of error bars. Journal of Climate, 18, 3699–3703. https://doi.org/10.1175/JCLI3499.1 Long, C. S., Fujiwara, M., Davis, S., Mitchell, D. M., & Wright, C. J. (2017). Climatology and interannual variability of dynamic variables in

multiple reanalyses evaluated by the SPARC Reanalysis Intercomparison Project (S-RIP). Atmospheric Chemistry and Physics, 17(23), 14,593–14,629. https://doi.org/10.5194/acp-17-14593-2017

Lu, J., Vecchi, G. A., & Reichler, T. (2007). Expansion of the Hadley cell under global warming. *Geophysical Research Letters*, 34, L06805. https://doi.org/10.1029/2006GL028443

Lucas, C., Timbal, B., & Nguyen, H. (2014). The expanding tropics: a critical assessment of the observational and modeling studies. *WIREs Climate Change*, *5*, 89–112. https://doi.org/10.1002/wcc.251

Nguyen, H., Evans, A., Lucas, C., Smith, I., & Timbal, B. (2013). The Hadley circulation in reanalyses: Climatology, variability, and change. Journal of Climate, 26(10), 3357–3376. https://doi.org/10.1175/JCLI-D-12-00224.1

Onogi, K., Tsutsui, J., Koide, H., Sakamoto, M., Kobayashi, S., Hatsushika, H., et al. (2007). The JRA25 Reanalysis. Journal of the Meteorological Society of Japan, 85(3), 369–432. https://doi.org/10.2151/jmsj.85.369

Quan, X., Hoerling, M. P., Perlwitz, J., Diaz, H. F., & Xu, T. (2014). How fast are the tropics expanding? Journal of Climate, 27, 1999–2013. https:// doi.org/10.1175/JCLI-D-13-00287.1

Rienecker, M. M., Suárez, M. J., Gelaro, R., Todling, R., Bacmeister, J., Liu, E., et al. (2011). MERRA: NASA's Modern-Era Retrospective Analysis for Research and Applications. *Journal of Climate*, 24(14), 3624–3648. https://doi.org/10.1175/JCLI-D-11-00015.1

Saha, S., Moorthi, S., H-L Pan, X. W., Wang, J., Nadiga, S., Tripp, P., et al. (2010). The NCEP climate forecast system reanalysis. Bulletin of the American Meteorological Society, 91(8), 1015–1058. https://doi.org/10.1175/2010BAMS3001.1

Seidel, D. J., Fu, Q., Randel, W. J., & Reichler, T. J. (2008). Widening of the tropical belt in a changing climate. *Nature Geoscience*, 1(1), 21–24. https://doi.org/10.1038/ngeo.2007.38

Stachnik, J. P., & Schumacher, C. (2011). A comparison of the Hadley circulation in modern reanalyses. Journal of Geophysical Research, 116, D22102. https://doi.org/10.1029/2011JD016677

Taylor, K. E., Ronald, S., & Meehl, G. (2012). An overview of CMIP5 and the experiment design. *Bulletin of the American Meteorological Society*, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1

Trenberth, K. E. (1991). Climate diagnostics from global analyses: Conservation of mass in ECMWF analyses. *Journal of Climate*, 4(7), 707–722. https://doi.org/10.1175/1520-0442(1991)004<0707:CDFGAC>2.0.CO;2

Trenberth, K. E., & Fasullo, J. T. (2018). Applications of an updated atmospheric energetics formulation. *Journal of Climate*, 31(16), 6263–6279. https://doi.org/10.1175/JCLI-D-17-0838.1

Trenberth, K. E., & Stepaniak, D. P. (2003). Seamless poleward atmospheric energy transports and implications for the Hadley circulation. *Journal of Climate*, *16*(22), 3706–3722. https://doi.org/10.1175/1520-0442(2003)016<3706:SPAETA>2.0.CO;2

Tselioudis, G., Lipat, B. R., Konsta, D., Grise, K. M., & Polvani, L. M. (2016). Midlatitude cloud shifts, their primary link to the Hadley cell, and their diverse radiative effects. *Geophysical Research Letters*, 43, 4594–4601. https://doi.org/10.1002/2016GL068242

Uppala, S. M., Kållberg, P. W., Simmons, A. J., Andrae, U., Da Costa Bechtold, V., Fiorino, M., et al. (2006). The ERA-40 re-analysis. *Quarterly Journal of the Royal Meteorological Society*, 131(612), 2961–3012. https://doi.org/10.1256/qj.04.176

Waliser, D. E., Shi, Z., Lanzante, J. R., & Oort, A. H. (1999). The Hadley circulation: Assessing NCEP/NCAR reanalysis and spare in-situ estimates. *Climate Dynamics*, 15(10), 719–735. https://doi.org/10.1007/s003820050312